Author’s response to reviews

Title: Risk factors and outcome of patients with eclampsia at a tertiary hospital in Egypt

Authors:

Ahmad Mahran (ezzeldin_ahmad@yahoo.com)
Hashem Fares (hashemfares1976@yahoo.com)
Reham Elkhateeb (rehamelkhateeb78@yahoo.com)
Mahmoud Ibrahim (hosnimahmoud60@yahoo.com)
Haitham Bahaa (haitham_bahaa@yahoo.com)
Ahmad Sanad (asasanad@hotmail.com)
Alaa Gamal (alaagamal316@gmail.com)
Mohamed Zeen El-Din (dr.muhamedzeen1980@gmail.com)
Eissa Khalifa (Eissa_mmk@yahoo.com)
Ahmed Abdelghany (Drahmed_abdlghany@yahoo.com)

Version: 4 Date: 01 Dec 2017

Author’s response to reviews:

Dear sir,

Thank you very much for potentially accepting my manuscript for publication. In response to editors and reviewers' comments:

1) Many thanks for highlighting changes in your text with yellow highlights. These are no longer required and we ask you to kindly upload a clean, final version of your manuscript without highlights or tracked changes.

- The highlights were removed from the manuscript.

2) Please remove all additional files not intended for publication, such as cover letters

- All additional files were removed.
3) Please include a statement at the end of the Authors contributions section to indicate that all authors have read and approved the final version of the manuscript.

- The required statement was added at the end of authors contributions section.